Author's response to reviews

Title: Inhibitory effects of prostaglandin E2 on collagen synthesis and cell proliferation in human stellate cells from pancreatic head adenocarcinoma.

Authors:

Ewa Pomianowska (ewa.pomianowska@medisin.uio.no)
Dagny Sandnes (d.l.sandnes@medisin.uio.no)
Krzysztof Grzyb (kgrzyb@ous-hf.no)
Aasa R Schjølberg (Aasa.R.Schjolberg@rr-research.no)
Monica Aasrum (monica.aasrum@medisin.uio.no)
Ingun H Tveteraas (i.h.tveteraas@medisin.uio.no)
Vegard Tjomsland (vegard.tjomsland@medisin.uio.no)
Thoralf Christoffersen (thoralf.christoffersen@medisin.uio.no)
Ivar P Gladhaug (i.p.gladhaug@medisin.uio.no)

Version: 7 Date: 8 May 2014

Author's response to reviews: see over
Re-submission MS 9031927179851771

Inhibitory effects of prostaglandin E$_2$ on collagen synthesis and cell proliferation in human stellate cells from pancreatic head adenocarcinoma

Dear Editor,

Thank you for your review. We have now revised the manuscript according to the editorial comment and a full statement on written informed consent from all patients is now included in the Methods section.

We hope that the manuscript will now be found suitable for publication in *BMC Cancer*.

Yours sincerely,

Ewa Pomianowska, MD.